Author's response to reviews

Title: Health workforce attrition in the public sector in Kenya: A look at the reasons

Authors:

Slavea G Chankova (slavea_chankova@abtassoc.com)
Stephen Muchiri (stephenMuchiri@yahoo.com)
Gilbert Kombe (gilbert_kombe@abtassoc.com)

Version: 2 Date: 24 May 2009

Author’s response to reviews:

We would like to thank the two peer reviewers for their comments and suggestions. We made revisions to the manuscript based on their input. All changes to the text are highlighted.

Revisions in response to review by Riita-Liisa Kolehmainen-Aitken: we made the suggested edits 1-4, and copy-edited all references (following the citation style required by the journal).

Revisions in response to review by Geoff King: we made revisions and additions to the manuscript to address the following specific points

(3) Added a comment in the first paragraph of the discussion section
(4) Added explanation in the first paragraph on p.5
(6) Added a comment in the second paragraph on p.8
(7) This is correct. Data on foreign workers was not collected so they were not included in the totals or in attrition numbers.

Thank you.